Author's response to reviews

Title: Delayed Recognition of Paediatric Calcaneal Osteomyelitis: a case report

Authors:

Alvin J Mallia (alvinmallia@hotmail.com)
Neil Ashwood (Neil.Ashwood@burtonft.nhs.uk)
Georgios Arealis (garealis@yahoo.gr)
Frank Bindi (Frank.Bindi@burtonft.nhs.uk)
Georgiana Zamfir (georgianazamfir@gmail.com)
Ilias Galanopoulos (galanop.ilias@gmail.com)

Version: 2
Date: 3 June 2015

Author's response to reviews:

Corrections made:

Please could you revise the CARE checklist. The page numbers of the given sections in your manuscript should be indicated. If the sections are not applicable with your submission, you may indicate N/A.

Please change the description of the patient’s gender to ‘male’.

Please include the Ethnicity of the patient in the Abstract and Case presentation sections.

As the patient is minor, please confirm if the consent statement was obtained from the parents for publication of this manuscript. We recommend the following wording is used for the consent section to be placed after the Conclusions section:

"Written informed consent was obtained from the patient’s legal guardian(s) for publication of this case report and any accompanying images. A copy of the written consent is available for review by the Editor-in-Chief of this journal."